Assessing internal variability of global mean surface temperature from observational data and implications for reaching key thresholds

Blair Trewin¹

¹Bureau of Meteorology (Australia)

November 26, 2022

Abstract

Observed global mean surface temperature (GMST) combines a forced response with internal variability, and quantifying internal variability is important in assessing the reaching of key thresholds, such as the 1.5 °C warming threshold in the Paris Agreement. This paper uses observational data to estimate internal variability. Since the current period of warming began in the 1970s, the 10-year mean of GMST has been very close to the 30-year mean for the period it is centred in and can therefore be considered as a robust indicator of the recent state of the climate. The range between the 5th and 95th percentile of annual residuals of observed GMST is 0.319 °C, substantially less than the corresponding range in large model ensembles, implying that the first individual year above 1.5 °C may occur later than indicated by climate models. The largest annual residuals are mostly associated with large-amplitude El Niño-Southern Oscillation (ENSO) events or major volcanic eruptions, with the relationship between cool years and La Niña more consistent than that between warm years and El Niño. The relationship between multi-year GMST means for differing periods indicates that the probability that the 1.5 °C threshold has been crossed (using the IPCC definition of the midpoint of the first 20-year period above the threshold) exceeds 50% once the most recent observed 11-year mean reaches 1.43 °C.

Hosted file

global temperature paper - agu template version.docx available at https://authorea.com/ users/546620/articles/606070-assessing-internal-variability-of-global-mean-surfacetemperature-from-observational-data-and-implications-for-reaching-key-thresholds

- 2 Assessing internal variability of global mean surface temperature from observational
- 3 data and implications for reaching key thresholds
- 4 Blair Trewin¹
- ⁵ ¹Australian Bureau of Meteorology, Melbourne, Australia
- 6 Corresponding author: Blair Trewin (<u>blair.trewin@bom.gov.au</u>)

7 Key Points:

- Observed interannual variability of global mean surface temperature is less than
 simulated in large model ensembles
- Observed variability can be used alongside wrecent data to estimate the probability that
 key temperature thresholds have been crossed
- 12

13 Abstract

- 14 Observed global mean surface temperature (GMST) combines a forced response with internal
- variability, and quantifying internal variability is important in assessing the reaching of key
- 16 thresholds, such as the 1.5 °C warming threshold in the Paris Agreement. This paper uses
- 17 observational data to estimate internal variability. Since the current period of warming began in
- the 1970s, the 10-year mean of GMST has been very close to the 30-year mean for the period it
- is centred in and can therefore be considered as a robust indicator of the recent state of the climate. The range between the 5th and 95th percentile of annual residuals of observed GMST is
- climate. The range between the 5^{th} and 95^{th} percentile of annual residuals of observed GMST is 0.319 °C, substantially less than the corresponding range in large model ensembles, implying
- 0.319 °C, substantially less than the corresponding range in large model ensembles, implying
 that the first individual year above 1.5 °C may occur later than indicated by climate models. The
- 23 largest annual residuals are mostly associated with large-amplitude El Niño-Southern Oscillation
- 24 (ENSO) events or major volcanic eruptions, with the relationship between cool years and La
- 25 Niña more consistent than that between warm years and El Niño. The relationship between
- ²⁶ multi-year GMST means for differing periods indicates that the probability that the 1.5 °C
- threshold has been crossed (using the IPCC definition of the midpoint of the first 20-year period
- above the threshold) exceeds 50% once the most recent observed 11-year mean reaches 1.43 °C.
- 29

30 Plain Language Summary

An increase in temperature of 1.5 °C from pre-industrial times is widely seen as a critical point in

climate change, but how do we know when warming of 1.5 °C has occurred? "Crossing" of 1.5

- ³³ °C is formally defined using 20-year averages, but since we don't want to wait many years to
- 34 know whether we've crossed 1.5 °C or not, how can we make use of the data we have now? This
- 35 paper shows how to use existing observations to determine how likely it is that 1.5 °C has been
- crossed. It also shows how much global temperatures vary from year to year, and when we might
 expect to start seeing individual years above 1.5 °C (which many people will interpret as
- expect to start seeing individual years above 1.5 °C (which many people will interpret as meaning that there has been a sustained crossing of 1.5 °C). Global temperatures do not vary as
- much from year to year in observations as they do in models, which suggests that the first years

40 above 1.5 °C may happen a little later than forecasts based on models say they will.

41 **1 Introduction**

42 Global mean surface temperature (GMST) is a key indicator used in the assessment of climate

- 43 change. The Paris Agreement has an objective of limiting the increase in global temperature to
- 44 well below 2 °C above pre-industrial levels, while pursuing efforts to limit the temperature
- 45 increase to 1.5 °C above pre-industrial levels.

- 47 The centrality of the 1.5 °C temperature threshold to the Paris Agreement, and climate policy
- 48 more generally, means that there is great public and policy interest in determining when it (and
- 49 other key thresholds) has been reached. This requires a metric which has a sufficient level of
- 50 stability not to be unduly influenced by short-term variability, whilst still being available in a
- reasonably timely manner. It is expected that individual years above 1.5 °C, which are likely to
- 52 be widely (if inaccurately) perceived as indicating a sustained breaching of the 1.5 °C threshold,
- ⁵³ will occur before longer-term metrics reach that level. (Individual months more than 1.5 °C

- above the 1850-1900 baseline have already occurred in all data sets used in this paper, in late
- 55 2015 and/or early 2016 at the peak of the 2015-2016 El Niño). The World Meteorological
- ⁵⁶ Organization (WMO) Annual to Decadal Climate Outlook issued in May 2021 (WMO, 2021)
- 57 stated a 40% probability that an individual year exceeding 1.5 °C would occur at some point in
- the period between 2021 and 2025, whilst projections reported (Arias et al., 2021; Lee et al., 2021) in the IPCC Sinth Assessment Penert (APC) indicate that for most emissions
- 2021) in the IPCC Sixth Assessment Report (AR6) indicate that for most emissions scenarios,
 the central estimate of the first 20-year period with a mean exceeding 1.5 °C has a midpoint in
- the early 2030s, with the probability of any individual year exceeding 1.5 °C reaching 40 to 60%
- by 2030 under most scenarios.
 - 63
 - 64 The volatility of short-term trends in global surface temperature has been well documented,
 - 65 particularly in the context of the relatively slow rate of global temperature increase between 1998
 - and 2012 (IPCC AR6 WGI Cross-Chapter Box 3.1; Eyring et al., 2021). Analysis using various
 - observational data sets by Liebmann et al. (2010), Risbey et al. (2014) and Marotzke and Forster
 - 68 (2015) all found that the difference between the most positive and most negative 15-year trends
 - in the observational record was in the range of $0.4 \,^{\circ}$ C to $0.5 \,^{\circ}$ C per decade. Less attention,
 - however, has been given to the variability of multi-year means and their stability as an indicator
 - 71 of the current state of the climate.

72 **2 Data and definitions**

- GMST is defined as the combination of air temperature at a nominal elevation of 2 m over land,
- and sea surface temperature over ocean. This is distinct from global surface air temperature
- 75 (GSAT) which is defined as air temperature at a nominal elevation of 2 m over both land and
- 76 ocean. The relationship between GMST and GSAT is discussed in Cross-Chapter Box 2.1 of the
- 77 Working Group I IPCC Sixth Assessment Report (Gulev et al., 2021). "Global surface
- temperature" is used as a generic term which covers both GMST and GSAT.
- 79
- 80 The GMST assessment in AR6 (Gulev et al., 2021) draws on four GMST data sets: HadCRUT5
- 81 (Morice et al., 2021), Berkeley Earth (Rohde and Hausfather, 2020), NOAAGlobalTemp-Interim
- (Vose et al., 2021) and Kadow et al (Kadow et al., 2020). These data sets meet the criteria of
- having data in at least 80% of the years between 1850 and 1900, data for at least 90% of global
- gridpoints in each year from 1960 onwards, and use the most recent generation of SST data sets
- (HadSST4 (Kennedy et al., 2019) or ERSSTv5 (Huang et al., 2017)) for their SST data. They can
- thus be considered to be globally quasi-complete since 1960 (addressing biases arising from
- 87 limited coverage of data-sparse areas at high latitudes in earlier generations of data sets), and to
- have sufficient temporal coverage to allow the calculation of a pre-industrial baseline. Annual
- and monthly mean temperature anomalies in each of the four data sets are calculated directly
- from the gridded data, with the global mean taken as the mean of the northern hemisphere and
- southern hemisphere means. This may differ from anomalies reported by the data providers
 themselves.
- 92 93
- 94 The primary metric used in AR6 for the assessment of observed changes in GMST is the
- 95 difference in mean GMST between 1850-1900 (taken as an observational approximation of the
- 96 pre-industrial period, which is not formally defined in the Paris Agreement) and the mean of the

97 most recent 10-year period (2011-2020). This difference is calculated independently for each of

- 98 the four component data sets and then averaged across the four data sets. Results are also
- reported for changes over a number of other time periods (e.g. from 1850-1900 to 1995-2014,
 which is the modern reference period used in CMIP6 model assessment). Linear trends are still
- which is the modern reference period used in CMIP6 model assessment). Linear trends are still
 reported in AR6 for a number of time periods but are no longer the primary metric for GMST
- assessment, as they had been in the Fifth Assessment Report and earlier IPCC reports. An annual
- (Figure 1) and monthly time series of anomalies from the 1850-1900 reference period is also
- 104 calculated by calculating anomalies for the four individual data sets and then averaging these.
- 105 These show sustained warming from about 1970 onwards; the period from 1970 to the present is
- 106 defined as the "current period of warming" in this paper.
- 107



108 109

Figure 1. GMST anomalies (1850-1900 baseline) from a mean of four data sets as described in the text.

- 112
- 113

The crossing time of a given global temperature threshold is defined in AR6 as the midpoint of the first 20-year period in which the global surface temperature exceeds that threshold (Arias et al., 2021). This is a different approach to that used in the IPCC Special Report on Global Warming of 1.5 °C (SR1.5) (Allen et al., 2018), which defines the global warming level at a given point of time as being the mean of the 30 year period control on that time, extrapolating

- given point of time as being the mean of the 30-year period centred on that time, extrapolating any secular trend into the future if necessary.
- 120

3 What length of averaging period is required to be a robust indicator of the state of theclimate?

- 123 In a changing climate, the choice of an averaging period to represent the recent climate is a
- balance between, on the one hand, having a period which is sufficiently long to capture major

- modes of internal variability in the global climate, and on the other hand, having a period which
- 126 excludes older data which are no longer representative of recent climate.
- 127
- 128 The question of optimal averaging periods for station data has been extensively considered in the
- literature (e.g. Angel et al. (1993), Huang et al. (1996), Srivastava et al. (2003)), with WMO
- 130 (2007) considering the question in a climate change context. 30 years has traditionally been
- 131 considered a standard averaging period, and is the period used by WMO to define a
- 132 climatological standard normal. WMO (2007) found, however, that for most variables (including
- temperature), the predictive accuracy of multi-year averages for periods from about 10-15 years
- upwards was not appreciably different from that of a 30-year average, suggesting that at the
- station level, a 10-15 year average is sufficient to capture most forms of internal variability.
- 136
- 137 One method of assessing the internal variability of *n*-year means relative to a 30-year mean
- 138 (taken here as a baseline) is to consider the variability of the residuals of the n-year means when
- 139 compared with the 30-year period with the same midpoint as the n-year period.
- 140



Figure 2. The standard deviation of the difference between global mean surface temperatures averaged over *n* years and the mean of the 30-year period that *n*-year period is entered in. The blue line shows the full record, and the red line the post-1970 period (using 30-year reference periods of 1971-2000 and later).

147

- 149 The results of this are shown in Figure 2, considering both the full GMST record and the period
- since the current sustained warming trend began in the 1970s. This indicates that, as expected,
- 151 the standard deviation of the residuals decreases as the period over which the means are taken
- increases. Over the full instrumental record, this is a steady decrease. However, from the 1970s

- onwards, the standard deviation of the residuals falls sharply, with increasing average period up 153
- to 8 years, but then much more slowly from 8 years thereafter. 154
- 155
- Considering in particular the 10-year averaging period used as part of the IPCC assessment of 156
- observed changes in global temperature, since 1970, the standard deviation of the residual is 157
- 0.019 °C (this compares with 0.013 °C for a 20-year period). The largest post-1970 difference 158
- between a 10-year mean and the 30-year mean it is centred in was for 1991-2000, where the 159
- 1991-2000 mean was 0.031 °C cooler than the 1981-2010 mean, in part reflecting the fact that 160
- post-1991 cooling due to the Pinatubo eruption had a larger influence on a mean taken over a 161
- shorter period. (The largest difference at any point in the record is 0.117 °C between the 1937-162
- 1946 and 1927-1956 means, reflecting the temporary reversal of global temperature trends 163
- around that time). This indicates that, unless there is a substantial change in the secular warming 164 trend, the 10-year average is sufficiently stable to be a relatively robust indicator of the current
- 165
 - state of the underlying climate. 166
 - 167

4 Internal variability in global surface temperature at the annual timescale 168

169 An estimate of internal variability in mean annual global temperatures is made by calculating the residual for each annual value (T_v for year y) from the mean of the 11-year period centred on 170 year y but excluding year y (i.e. $T_{y-5},...,T_{y-1}, T_{y+1},...,T_{y+5}$). The 11-year period is chosen to match 171

as closely as possible the 10-year period used for IPCC temperature assessments whilst allowing 172

- 173 a period which is symmetric about the candidate year.
- 174

175 Figure 3 shows the residuals in each individual year and the frequency distribution of the

residuals, while Table 1 shows the 5-95% percentile range in the residuals (corresponding to the 176

IPCC 'very likely' range), both for the combined time series and for the four individual 177

- component data sets. The annual residuals are weakly but not significantly autocorrelated (r =178 179 +0.234).
- 180



181 182

Figure 3. (Left) Time series of the annual global temperature residuals for the combined time 183 series. (Right) Frequency distribution of these residuals (labels show the midpoint of alternate 184 0.02 °C intervals). 185

95th percentile

0.180

0.170

0.164

0.190

0.173

 $(95^{th} - 5^{th})$

percentile

0.337

0.318

0.304

0.353

0.319

187

188 **Table 1**

NOAA

Combined

Berkeley Earth

189 Standard deviation and 5-95 percentile range of annual residuals ($^{\circ}C$)

190 191

Data setStandard
deviation5th percentileHadCRUT50.102-0.157Kadow0.099-0.148

0.093

0.108

0.099

192

193

194 This shows that the width of the 'very likely' (5-95%) range for the combined data set is 0.319

-0.140

-0.163

-0.146

¹⁹⁵ °C. There is little difference between the value for the combined data set and that in the four

individual component data sets (which range from 0.304 °C to 0.353 °C), indicating that the

197 process of averaging the four data sets does not substantially smooth interannual variability. By

way of comparison, the width of the 5-95% range in various large CMIP6 model ensembles reported in AB6 (Lee et al. 2021) is between 0.4 °C and 0.6 °C, indicating that these models

reported in AR6 (Lee et al., 2021) is between 0.4 °C and 0.6 °C, indicating that these models are simulating more interannual variability than is occurring in observational data sets. This is also

201 consistent with the results which Bengtsson and Hodges (2019) obtained using the HadCRUT4 202 data set (an earlier generation than the HadCRUT5 data set used in this analysis).

203

The largest individual positive residuals are associated with strong El Niño events, in 1878

(+0.280 °C), 1915 (+0.227 °C) and 1889 (+0.220 °C), consistent with the known relationship

between the El Niño-Southern Oscillation (ENSO) and global surface temperatures (e.g.

Trenberth et al. (2002), Thompson et al. (2009)). Recent reassessments of the 1877-1878 El Niño

event (Huang et al. (2020), Vaccaro et al. (2021)) indicate that in terms of Pacific SST

anomalies, it was of comparable intensity to the well-known extreme El Niño events of 19821983, 1997-1998 and 2015-2016. 1998 had the largest annual mean temperature residual (+0.191

°C) of the post-1915 period. The temperature signal of the 1982-1983 El Niño was largely offset

by the 1982 eruption of El Chichón. (A final residual for 2016 cannot yet be calculated using this

methodology but data available to date indicate that it is likely to be in the order of 0.16 °C to

214 0.17 °C).

The largest negative residuals are associated either with strong La Niña events (1976 (-0.212)

²¹⁷ °C), 1917 (-0.205 °C), 1956 (-0.196 °C)) or major volcanic eruptions (1964 (-0.200 °C), 1904

218 (-0.178 °C)). Although the 1991 Pinatubo eruption had the largest negative radiative forcing of

any post-1900 eruption (Luo (2018), Gulev et al. (2021)), El Niño conditions existed for much of the 1001 05 period, partly offsetting the velocities signal, and haves the residual in this

the 1991-95 period, partly offsetting the volcanic signal, and hence the negative residuals in this data associated with Pinatubo are smaller than those associated with Mount Agung (1963) or the

1902 eruptions, with a lowest value of -0.157 °C in 1992. Although uncertainties in individual

annual values are substantially greater pre-1900 as a result of sparser observations, there is little

evidence of any signal in the observed interannual variability, with the standard deviation of the

residuals over the full period of record (0.099 °C) similar to that for the post-1900 period (0.100 $^{\circ}$ C).

227



228 229

Figure 4. Annual residuals of global mean surface temperature and values of the Niño 3.4 index
(ERSSTv5, detrended, mean of September-November in the preceding year). The three years
following each of the eruptions of Mount Agung (1963), El Chichón (1982) and Pinatubo (1991)
are shown in black. Data shown are from 1950-51 to 2014-15.

234

235

The ENSO influence on global temperatures is further illustrated in Figure 4, which shows GMST residuals as a function of the value of the Niño 3.4 index in September-November of the preceding year, from 1950 when reliable values of the index commenced. The relationship is especially consistent in La Niña years; excluding years affected by volcanic eruptions, 9 of the 10 years with the coolest GMST residuals make up 9 of the 10 years with the largest negative values of the Niño 3.4 index. The relationship between El Niño and warmth is less consistent;

values of the Niño 3.4 index. The relationship between El Niño and warmth is less consistent;
 while the three strongest non-volcanic El Niño years all have GMST residuals exceeding +0.1

°C, anomalously warm years have also occurred under neutral ENSO conditions, with 1953 and

- 1980 both exceeding +0.15 °C.
- 245

To explore the relationship further, residuals were recalculated using GMST for 12-month

247 periods starting in each of the 12 calendar months, rather than just the calendar year. The

correlation of these with the September-November Niño 3.4 values is shown in Figure 5, and

- shows that the ENSO influence on 12-month means of GMST manifests most strongly for 12-
- month means beginning in months between August and November, and most weakly for 12-
- month means beginning in March and April. However, this has little impact on the overall
- variability of 12-month mean temperatures, with the standard deviation of 12-month residuals
- ranging between 0.097 $^{\circ}$ C and 0.103 $^{\circ}$ C regardless of the start month.





Figure 5. (Red) Correlation between 12-month mean GMST for periods starting in the stated month, and the Niño 3.4 index (as per Figure 4, using Niño 3.4 for the preceding year for GMST 12-month periods starting in January to March). (Blue) Standard deviation (°C) of residuals of the 12-month mean GMST for 12-month periods starting in the stated month.

263 **5 Probability of individual years above thresholds**

Table 2 shows the crossing times of thresholds which have already been crossed, along with the 264 first individual year above the threshold. Since the present phase of consistent warming began in 265 the 1970s, the first year above a threshold has generally occurred between 1 and 6 years before 266 the midpoint of the crossing period for that threshold. The exception is the 0.9 °C threshold, 267 which was reached (with 0.901 °C) in 1998, 8.5 years before the midpoint of the 1997-2016 268 269 crossing period, indicating the sensitivity of such an analysis to the specific thresholds used, as the next year above 0.9 °C did not occur until 2005. (Conversely, 2010 reached 0.996 °C, four 270 years before the 1.0 °C threshold was reached for the first time). 271 272

273 Table 2

- Crossing periods for particular temperature thresholds and the first individual year in which
 those thresholds were reached
- 276

Threshold (°C)	Crossing period	First year above threshold	Difference with midpoint of crossing period (years)
0.3	1964-1983	1878	96.5

0.4	1971-1990	1944	36.5
0.5	1976-1995	1980	5.5
0.6	1982-2001	1988	3.5
0.7	1987-2006	1995	1.5
0.8	1993-2012	1998	4.5
0.9	1997-2016	1998	8.5
1.0	Not yet occurred	2014	
1.1	Not yet occurred	2015	
1.2	Not yet occurred	2016	

278

The probability of an annual residual above +0.12 °C in the observed data is approximately 10%

(Figure 6). The central estimates of crossing times projected in AR6 imply warming rates for the
period from 2001-2020 to the date of reaching the 1.5 °C threshold which range from 0.021
°C/year (SSP1-1.9) to 0.030 °C (SSP5-8.5) (Table 3). Assuming that this rate of secular warming

is constant over that period, this implies crossing of a 1.38 °C threshold, and hence a probability

of an individual year exceeding 1.5 °C reaching 10%, at a point 4 to 6 years before the midpoint of the crossing period, and thus around 2025 for SSP5-8.5 and 2029 for SSP1-1.9. This is broadly consistent with the observations reported in Table 2 for lower thresholds. It can also be compared with the WMO outlook for 2021-2025 issued in May 2021, which assessed a 40% probability of an individual year exceeding 1.5 °C in that period (which, assuming a constant

secular warming trend over that period, equates roughly to a 10% probability in the midpointyear, 2023).

291

292 **Table 3**

293 Central estimate of IPCC AR6 1.5 °C crossing period, implied warming rate and implied

294 crossing of 1.38 °C threshold

295

Scenario	Central estimate of 1.5 °C crossing	Implied warming rate from 2001-2020	Projected crossing of 1.38 °C threshold
	period ^a	(°C per year)	
SSP1-1.9	2025-2044	0.021	2029
SSP1-2.6	2023-2042	0.023	2027
SSP2-4.5	2021-2040	0.025	2026
SSP3-7.0	2021-2040	0.025	2026
SSP5-8.5	2018-2037	0.030	2025

296

297 Note: Implied warming rate is from 2001-2020 to the stated 1.5 °C crossing period. Projected

crossing of 1.38 °C assumes a constant warming rate from 2001-2020 to 1.5 °C crossing.

^a Arias et al. (2021)





305 6 Assessing the likelihood that a threshold has been crossed from observational data

306 Using the 20-year period used for the crossing time definition in AR6, it can only be determined definitively that a threshold has been crossed 10 years after the event. However, in a consistently 307 warming climate, well before that period is completed the probability that the threshold has been 308 crossed will be very close to 1. To give one example, as of the end of 2020, there had not yet 309 been a 20-year period with mean temperatures more than 1.0 °C above those of 1850-1900 (the 310 2001-2020 mean anomaly was +0.99 °C), and hence a definitive crossing date for the 1.0 °C 311 cannot yet be determined; however, for crossing to have occurred before 2020, a sufficient 312 condition is for the 2010-2029 mean to be above 1.0 °C, which would require a 2021-2029 mean 313 above 0.9 °C. As this value is 0.19 °C below the 2011-2020 mean, and lower than any individual 314 year from 2012 onwards, it was extremely likely as of the end of 2020 that crossing of the 1.0 °C 315 threshold had occurred before 2020. 316

317

The purpose of this section is to establish an objective method of assessing the probability that

319 crossing has occurred by a certain year, as a function of the available observational data and 320 historical probability distributions. Figure 7 shows indicators of the distribution of the difference

between 20-year means of GMST commencing in year y, and *n*-year means of GMST

commencing in year y, since the current warming phase began in the 1970s. This allows the

probability that a 20-year period projecting into the future will have GMST exceeding a certain

threshold to be estimated, as a function of an *n*-year observed mean ending in the most recent

325 year.

- 327 Of particular interest here is the case of n = 11, since if the 20-year mean for the period
- extending from year (yy 10) to (yy + 9) exceeds a threshold, that means that threshold has been

- 329 crossed on or before year *yy*. For n = 11, the 10th percentile of the differences is 0.060 °C, the
- 50^{th} percentile is 0.079 °C, and the 90th percentile is 0.101 °C. This indicates that, providing
- there is no major change in the underlying warming rate, the probability that 1.5 °C has been crossed on or before the current year exceeds 50% once the observed mean for the most recent
- 11 years reaches 1.43 °C. The probability is about 90% once the 11-year mean reaches 1.44 °C,
- while crossing is unlikely to have occurred if the observed 11-year mean is below 1.40 °C. There
- is no instance in the post-1970 observed data of a difference of less than $0.05 \,^{\circ}\text{C}$ between the 11-
- and 20-year means (Figure 7, right).
- 337



Figure 7. (Left) Differences in observed data (1971-2020) between the 20-year mean of GMST starting in year *y* and the *n*-year mean starting in year *y*. (Right) Frequency distribution of the differences (°C) between the 20- and 11-year means.

343 344

345 7 Discussion

The results obtained indicate that interannual variability in observed GMST is lower than that

347 generally observed in model ensembles, which may indicate that models are overestimating

348 global temperature variability, that observational data are not capturing the full range of

variability which occurs, or a combination of the two. A possible scenario for observational data

under-representing variability is that the methods used to interpolate GMST data sets over data-

sparse areas fails to fully capture interannual variability in those areas; however, Bengtsson and Hodges (2019)'s finding that interannual variability was lower in observations than models still

held when they masked models to the areas with available data in the HadCRUT4 data set, which

held when they masked models to the areas with available data in the HadCRUT4 da
 is not interpolated and treats data voids as missing.

355

356 While a full consideration of why models might overestimate interannual variability of global

surface temperature is beyond the scope of the paper, it has been found (Grose et al., 2020) that

358 CMIP6 models tend to overestimate SST variability in most of the central and eastern equatorial

359 Pacific (and hence overestimate ENSO amplitude), whilst they also tend to overestimate the

duration of El Niño events (Figure 3.36, Eyring et al. 2021). As a substantial proportion of

interannual GMST variability is attributable to ENSO, it would be expected that the capacity of a

model to simulate ENSO amplitude, and the relationships between ENSO and GMST, would be
 important in determining its capacity to accurately simulate interannual variability of global
 surface temperature.

364 365

The results obtained here also indicate that, while trends of global surface temperature over short periods (10-15 years) are volatile, means of global temperature over a similar length of time are a relatively robust indicator of the recent state of the climate. This allows them to be used to assess the likelihood that key thresholds have been crossed with a relatively high level of confidence. This will become increasingly important as the world approaches 1.5 °C of warming since the pre-industrial period, which is likely to occur during the late 2020s or early 2030s under all emission scenarios.

- 373
- 374

375 Acknowledgments

Comments from Greg Browning and David Jones on earlier drafts of the manuscript are greatlyappreciated.

- 378
- This work was funded by the Australian Bureau of Meteorology.
- 380

381 **Open Research**

- The annual global mean surface temperature (GMST) time series used in this paper are available
- at <u>https://doi.org/10.5281/zenodo.6321535</u>. (Note for editor/reviewers: it is expected that by the
- time of publication, it will also be available in the IPCC AR6 data repository, in which case that
- reference will be added).

386

387

388 **References**

- Allen, M.R., Dube, O.P., Solecki, W., Aragón-Durand, F., Cramer, W., Humphreys, S., et al.
- 390 (2018). Framing and Context. In: *Global Warming of 1.5°C. An IPCC Special Report on the*
- impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse
- 392 gas emission pathways, in the context of strengthening the global response to the threat of
- 393 climate change, sustainable development, and efforts to eradicate poverty [Masson-Delmotte,V.,
- P.Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan,
- R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T.
- 396 Maycock, M. Tignor, & T. Waterfield (eds.)]. Geneva: World Meteorological Organization.

- Angel, J.R., Easterling, W.R., & S.W. Kirtsch. (1993). Towards defining appropriate averaging
 periods for climate normals. *Climatological Bulletin*, 27, 29-44.
- Arias, P.A., Bellouin, N., Coppola, E., Jones, R.G., Krinner, G., Marotzke, J., et al. (2021).
- 400 Technical Summary. In: Climate Change 2021: The Physical Science Basis. Contribution of
- 401 Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate
- 402 Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, N. Caud,
- 403 Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T. K.
- Maycock, T. Waterfield, O. Yelekçi, R. Yu, & B. Zhou (eds.)]. Cambridge University Press. In
 Press.
- Bengtsson, L., & Hodges, K.I. (2019). Can an ensemble climate simulation be used to separate
 climate change signals from internal unforced variability? *Climate Dynamics*, *52*, 3553-3573,
 https://doi.org/10.1007/s00382-018-4343-8
- 409 Eyring, V., Gillett, N.P., Achuta Rao, K.M., Barimalala, R., Barreiro Parrillo, M., Bellouin, N.,
- 410 et al. (2021). Human Influence on the Climate System. In: *Climate Change 2021: The Physical*
- 411 Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the
- 412 Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L.
- 413 Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K.
- Leitzell, E. Lonnoy, J. B. R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, & B.
- 415 Zhou (eds.)]. Cambridge University Press. In Press.
- 416 Grose, M.R., Narsey, S., Delage, F.P., Dowdy, A.J., Bador, M., Boschat, G., et al. (2020).
- Insights from CMIP6 for Australia's future climate. *Earth's Future*, 8, e2019EF001469,
- 418 https://doi.org/10.1029/2019EF001469
- 419 Gulev, S. K., Thorne, P.W., Ahn, J., Dentener, F.J., Domingues, C.M., Gerland, S., et al. (2021).
- 420 Changing State of the Climate System. In: *Climate Change 2021: The Physical Science Basis.*
- 421 Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel
- 422 on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger,
- 423 N. Caud, Y. Chen, L., Goldfarb, M. I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J. B. R.
- Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu and B. Zhou (eds.)]. Cambridge
 University Press. In Press.
- Huang, B., Thorne, P.W., Banzon, V.F., Boyer, T., Chepurin, G., Lawrimore, J.H., et al. (2017).
- 427 Extended Reconstructed Sea Surface Temperature, version 5 (ERSSTv5): upgrades, validations,
- 428 and intercomparisons. *Journal of Climate*, *30*, 8179-8205, https://doi.org/10.1175/jcli-d-16-429 0836.1
- Huang, B., L'Heureux, M., Hu, Z.-Z., Yin, X. & Zhang, H. -M. (2020). How significant was the
- 431 1877/78 El Niño? Journal of Climate, 33, 4853-4869, https://doi.org/10.1175/jcli-d-19-0650.1
- Huang, J., van den Dool, H.M. & Barnston, A.G. (1996). Long-lead seasonal temperature
- 433 prediction using optimal climate normal. *Journal of Climate*, 9, 809-817.
- 434 Kadow, C., Hall, D.M. and Ulbrich, U. (2020). Artificial intelligence reconstructs missing
- climate information. *Nature Geoscience*, *13*, 408-413, https://doi.org/10.1038/s41561-020-05825
- 437 Kennedy, J.J., Rayner, N.A., Atkinson, C.P. & Killick, R.E. (2019). An ensemble data set of sea-
- 438 surface temperature change from 1850: the Met Office Hadley Centre HadSST.4.0.0.0 data set.
- 439 Journal of Geophysical Research: Atmospheres, 124, 7719-7763,
- 440 http://doi.org/10.1029/2018jd029867
- Lee, J. Y., Marotzke, J., Bala, G., Cao, L., Corti, S., Dunne, J.P., et al., (2021). Future Global
- 442 Climate: Scenario-Based Projections and Near-Term Information. In: Climate Change 2021: The

- 443 Physical Science Basis.Contribution of Working Group I to the Sixth Assessment Report of the
- 444 Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L.
- 445 Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K.
- Leitzell, E. Lonnoy, J. B. R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu and B.
- 447 Zhou (eds.)]. Cambridge University Press. In Press.
- Liebmann, B., Dole, R.M., Jones, C., Bladé. I. & Allured, D. (2010). Influence of choice of time
- 449 period on global surface temperature trend estimates. *Bulletin of the American Meteorological*
- 450 Society, 91, 1485-1492, https://doi.org/10.1175/2010BAMS3030.1
- Luo, B. (2018). Aerosol Radiative Forcing and SAD version v4.0.0 1850–2016. Retrieved from:
 ftp://iacftp.ethz.ch/pub_read/luo/cmip6_sad_radforcing_v4.0.0.
- 453 Marotzke, J. & Forster, P.. 2015. Forcing, feedback and internal variability in global temperature
- 454 trends. *Nature*, *517*, 565-570, https://doi.org/10.1038/nature14117
- 455 Morice, C.P., Kennedy, J.J., Rayner, N.A., Winn, J.P., Hogan, E., Killick, R.E., et al. (2021). An
- updated assessment of near-surface temperature change from 1850: the HadCRUT5 dataset.
- 457 Journal of Geophysical Research: Atmospheres, 126, https://doi.org/10.1029/2019jd032361
- 458 Risbey, J., Lewandowsky, S., Langlais, C., Monselesan, D.P., O'Kane, T.J. & Oreskes, N.
- (2014). Well-estimated global surface warming in climate projections selected for ENSO phase.
- 460 *Nature Climate Change*, *4*, 835-840, https://doi.org/10.1038/nclimate2310
- Rohde, R.A. & Hausfather, Z. (2020). The Berkeley Earth Land/Ocean temperature record.
- 462 *Earth Systems Science Data*, *12*, 3469-3479, https://doi.org/10.5194/essd-12-3469-2020
- 463 Srivastava, A.K., Guhathakurta, P. & Kshirsagar, S.R. (2003). Estimation of annual and seasonal
- temperatures over Indian stations using optimal normal. *Mausam*, 54, 615-622.
- Thompson, D.W.J, Wallace, J.M., Jones, P.D. & Kennedy, J.J. (2009). Identifying signatures of
- natural climate variability in time series of global-mean surface temperature: methodology and
- 467 insights. *Journal of Climate*, 22, 6120-6141, https://doi.org/10.1175/2009JCLI3089.1
- Trenberth, K.E., Caron, J.M., Stepaniak, D.P. & Worley, S. (2002). Evolution of El Niño-
- 469 Southern Oscillation and global atmospheric surface temperatures. *Journal of Geophysical*
- 470 *Research*, *107* (*D*8), https://doi.org/10.1029/2000JD000298
- 471 Vaccaro, A., Emile-Geay, J., Guillot, D., Verna, R., Morice, C., Kennedy, J. & Rajaratnam, B.
- 472 (2021). Climate field completion via Markov random fields application to the HadCRUT4.6
- 473 temperature dataset. Journal of Climate, 34, 4169-4188, https://doi.org/10.1175/jcli-d-19-0814.1
- Vose, R.S., Huang, B., Yin, X., Arndt, D., Easterling, D.R., Lawrimore, J.H., et al. (2021).
- 475 Implementing full spatial coverage in NOAA's global temperature analysis. *Geophysical*
- 476 *Research Letters*, 48, e2020GL090873, https://doi.org/10.1029/2020gl090873
- 477 World Meteorological Organization (2007). The role of climatological normals in a changing
- 478 climate. *WCDMP No. 61*, World Meteorological Organization, Geneva, 130 pp.
- 479 World Meteorological Organization (2021). WMO Global Annual to Decadal Climate Update.
- 480 <u>https://hadleyserver.metoffice.gov.uk/wmolc/WMO_GADCU_2020.pdf</u>, published 27 May
- 481 2021, viewed 5 January 2022.
- 482